Author's response to reviews

Title: Racial disparities in infant mortality: What has birth weight got to do with it and how large is it?

Authors:

Timothy B Gage (tbg97@albany.edu)
Fu Fang (ff2393@albany.edu)
Erin O'Neill (eo0379@albany.edu)
A. Gregory DiRienzo (gdirienzo@albany.edu)

Version: 4 Date: 2 December 2010

Author's response to reviews: see over
To the Editors of BMC

This is a resubmission of the manuscript entitled “Racial disparities in infant mortality: What has birth weight got to do with it and how large is it?”.

   a) We have made all of the changes (essential and discretionary) recommended by the latest review.
   b) We have copy edited the paper for typos and clarity.
   c) We have checked the manuscript for format using the check list indicated in your most recent email.

If any additional corrections are required please let me know. I agreed to publication costs when I submitted this paper. I assume that information on how to pay you will either be on the submission website or that you will email me an invoice of some sort.

Sincerely

Timothy B. Gage